Author's response to reviews

Title: Gender differences in the association between socioeconomic status and hypertension incidence: the Korean Genome and Epidemiology Study

Authors:

Tae-Hwa Baek (taehwa.baek@gmail.com)
Hae-Young Lee (hylee612@snu.ac.kr)
Nam-Kyoo Lim (namkyoo.lim@gmail.com)
Hyun-Young Park (mdhypark@gmail.com)

Version: 2 Date: 26 March 2015

Author's response to reviews: see over
We are pleased to submit an original article entitled “Gender differences in the association between socioeconomic status and hypertension incidence: the Korean Genome and Epidemiology Study (KoGES)” by Baek TH et al. for consideration for publication in the BMC Public Health. In this manuscript, we report the results of a prospective study that low socioeconomic status is associated with an increasing risk of incidence hypertension and there are gender differences in the relationship between SES and the incidence of hypertension in Korean adults. We believe that our findings could be of interest to your readers.

This manuscript has not been published and is not under consideration for publication elsewhere. We have no conflicts of interest to disclose.

Thank you for your kind considerations in advance and we are looking forward to hearing a positive reply from you soon.

Sincerely yours,

Hyun-Young Park, MD, PHD
Head of Division, Division of Cardiovascular and Rare Disease
Korea National Institute of Health